2006-03-16 Pbar Department Meeting

Department Head: Keith Gollwitzer

Machine Conditions:

Shutdown:

Shutdown Progress
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Motor. Kicker Stand
5% done

Week 3
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Quad BPM
5% done

4-8GHz Core
0% done
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AP50 SO cord replacement
75% done

AP2 Vacuum gate
0% done
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Install AP2 Trims
95% done

PBAR Shutdown Status Mar 10, 2006

Install new Beam Stop
0% done

Rotate AP2 BPM
0% done

AP30 Tank Leak
0% done

=3}

AP30

Thursday, March 16, 2006

New D10 Pipe
50% done

Debuncher
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Acc BPMs
0% done

Deb Quad Shunts.
1% done

Upgrade Deb. BPM
50% done

LM Calib.
0% done

APOD

Install New Target

0% done
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LCW Flush || Survey Level Run || Feeder 24
0% done 20% done 70% done
AP1 PS

0% done

Install Hall Probes
0% done

Install OTR
0% done J/

APO SO cord replacement

0% done

Beam Line BPM Cable
0% done

Shutdown status
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Motor kicker stand -

Pipe in D10 back under vacuum

Quad BPMs - normal one pulled out.

reorganization of instrumentation.
New trim PS in A10 - need 480V

AP50 SO cord - 100%
APO SO cord - 50% done

4-8GHz core - Ralph and Co. have measured all cables at least 4 times.

fiber optic cable gives us a few nsec.
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The skew Quad BPM slid downstream.

9:59 AM
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fiber optic cable gives us a few nsec.
AP30 tanks. Henry expected small leak in vacuum. Pressurized.
Debuncher quad shunts. Need 13 cards. They are not all done. 10%
Debuncher BPM - one more sector to go.
BLM calibration - 10 and 60 checked. Labels and cables. 2%
AP1 power supply switch - Still have to swap the cables between 105 and 109.
All power supplies have had their fans replaced.
Hall probes.
OTR - Not due for installation until April.
Testing of applications in the Linux world. Let Wally know which ones you want to
test.
= Make sure it initializes
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